Determinants of capital structure: evidence from a major developing economy

Bülent Köksal · Cüneyt Orman

Accepted: 23 June 2014/Published online: 22 July 2014 © Springer Science+Business Media New York 2014

Abstract A major shortcoming of capital structure studies on developing economies is that they generally restrict their analyses to large publicly-traded manufacturing firms. Consequently, we know little about the applicability of various capital structure theories to firms that are private, small, and/or outside the manufacturing industry in these economies. In this paper, we conduct a comparative test of the trade-off and pecking order theories using a comprehensive firm-level dataset that covers manufacturing, nonmanufacturing, small, large, publicly-traded, and private firms in a major developing economy, Turkey. The trade-off theory provides a better description of the capital structures of all firm types than the pecking order theory. Moreover, the trade-off theory appears to be particularly suitable for understanding the financing choices of large private firms in the non-manufacturing sector and when the economic environment is relatively stable. By contrast, pecking order theory is most useful when it comes to small publicly-traded manufacturing firms, especially when the economic environment is relatively unstable.

B. Köksal

Department of Economics, Ipek University, Çankaya, 06550 Ankara, Turkey

C. Orman (🖂)

Central Bank of the Republic of Turkey, Istiklal Cad., No: 10, Ulus, 06100 Ankara, Turkey e-mail: cuneyt.orman@tcmb.gov.tr **Keywords** Capital structure \cdot Non-financial firms \cdot Pecking order theory \cdot Trade-off theory \cdot Turkey

JEL Classification G30 · G32 · L26

1 Introduction

Ever since the publication of Modigliani and Miller (1958)'s "irrelevance proposition", there has been extensive theoretical work on the determinants of firms' capital structures. Already by the early 1980s, these efforts culminated in the development of the two major theories of capital structure. In the (*static*) *trade-off theory*, firms trade off tax savings from debt financing against deadweight costs of possible bank-ruptcy. The *pecking order theory*, on the other hand, posits that, due to adverse selection, firms prefer internal to external financing and debt to equity if external financing is used. Even though neither theory is completely satisfactory, they have been instrumental in identifying many of the factors that govern firms' actual financing decisions.

Beginning with the mid-1980s, there has been an outpouring of empirical research aimed at comparing and contrasting the predictive powers of the major theories of capital structure, often formulated as a contest between the trade-off theory and the pecking order theory. While the earlier work concentrated on the developed economies, particularly the United States, a major preoccupation of recent research has been to test these theories in the context of developing economies. The evidence coming from both the developed and developing economy capital structure studies thus far is decidedly mixed; while some studies provide support for the trade-off theory, others lend support for the pecking order theory. Often, studies report conflicting findings even for a given country. In the quest for resolving the contest between the tradeoff and pecking order theories, researchers have subsequently taken a variety of routes that include modifying and/or improving the existing theories, developing new theories, and using better data and/or methods. Despite considerable progress, the issue is yet to be settled, both for developed and developing economies, and more so for the latter.

A major shortcoming in developing economy capital structure studies, including Turkey, is that they generally restrict their analyses to manufacturing firms listed on the stock exchanges, which incidentally tend to be relatively large firms. As such, these studies cannot speak to the capital structure choice of the average (or typical) non-financial firm, nor do they inform us about the capital structure choices of firms that are privately-held, firms outside the manufacturing industry, or firms that are relatively small. This also implies that, by design, comparative tests of the theory in such studies can be informative only in the specific context of large public manufacturing firms.

In this paper, we contribute to the capital structure literature on developing economies by investigating the capital structure choices of Turkish non-financial firms and also by doing a comparative test of the tradeoff and pecking order theories. Our study differs from previous studies in that we utilize a unique and comprehensive dataset compiled by the Central Bank of the Republic of Turkey (CBRT) that provides financial information on small, large, public, private, manufacturing as well as non-manufacturing nonfinancial firms in Turkey for the past 20 years. The fact that our dataset is substantially more comprehensive in terms of firm types as well as leverage determinants suggested by theory compared to the datasets used in previous studies enables us to conduct a number of systematic analyses which, to our knowledge, are new in the context of capital structure research on developing economies. Specifically, since our dataset is considerably more representative of the universe of Turkish non-financial firms, it enables us to take a much more accurate picture of the capital structure of the average Turkish non-financial firm. It also allows us to investigate systematically the capital structure differences between firms of various kinds: manufacturing versus non-manufacturing firms, small versus large firms, and public versus private firms. Accordingly, we are able to carry out a considerably more comprehensive comparative test of the trade-off and pecking order theories than previous studies.

The extant literature reviewed in Sect. 2 has identified several factors that are correlated with firm leverage in both developed and developing countries. Following this body of work, we investigate concurrently the influences on leverage of four broad types of variables: Firm-specific, tax-related, industry-specific, and macroeconomic. Several results emerge from our empirical analysis. In particular, we find that, for the average Turkish non-financial firm, most of the independent variables have the signs that would be expected in light of previous theoretical work and the empirical findings obtained for other countries. Specifically, we provide evidence that leverage (shortterm, long-term, and total) is positively correlated with firm size, potential debt tax shields, industry median debt ratios, and inflation, and negatively correlated with profitability, business risk, and real GDP growth. Asset tangibility is positively correlated with longterm and total leverage but negatively correlated with short-term leverage. Firm growth, on the other hand, does not seem to be related with firms' leverage decisions. Our findings, therefore, confirm, from a capital structure point of view, the statistical relevance of tax-related, industry-related, and macroeconomic factors in addition to firm-level factors in the context of Turkish non-financial firms.

In addition to statistical significance, we analyze the economic significance of leverage determinants, a first in the capital structure literature on developing economies. Our findings indicate that each of the four broad types of leverage determinants does indeed play an economically important role in shaping the capital structure decisions of Turkish non-financial firms during our sample period. Specifically, we find that while potential debt tax shields are the most economically significant determinant of short-term and total leverage, tangibility is the most economically significant determinant of long-term leverage. Firm size is another major determinant of leverage, second only to tangibility and potential debt tax shields for long-term and total leverage, respectively. Industry median leverage and inflation also play a notable but smaller role when compared to firm size, profitability, tangibility, and tax-related determinants.

We next carry out a rare exercise in the literature on corporate capital structure in developing economies. Specifically, we investigate the capital structure differences between manufacturing and nonmanufacturing, large and small, and public and private firms. One of our main findings is that regardless of their size, industrial membership, and stock market listing, firms' short-term leverage is much larger than their long-term leverage. This indicates that short-term debt finance was and still is the norm for Turkish non-financial firms, which is not desirable both from economic efficiency and financial stability perspectives. We also find that while manufacturing firms have higher leverage (both short-term and long-term) than non-manufacturing firms, large firms and public firms have higher long-term leverage but lower short-term leverage than small firms and private firms, respectively. This implies that small private manufacturing firms are the firms with the greatest proportion of short-term debt in their capital structures. In addition to these differences in leverage patterns, firm types also differ in terms of how various determinants are related with leverage. The most significant differences appear to be neither between public and private firms nor between manufacturing and nonmanufacturing firms but rather between small and large firms.

Another novel aspect of our paper is that we explore the implications of international capital flows for firms' capital structures. Studying the capital structure implications of capital flows is interesting because capital flows exert, either directly or via domestic capital markets, a major influence on the availability and allocation of funds to various types of firms in many developing economies, including Turkey. Moreover, both the size and volatility of capital flows grew rapidly during the past two decades, particularly after the early 2000s, raising concerns over economic and financial stability in policy circles. Our results do in fact show that there is a strong positive correlation between capital flows and corporate leverage in Turkey, indicating that firms choose to intensify their usage of debt in response to a rise in the availability of foreign capital. Moreover, the debt-promoting effect of capital flows appears to be particularly strong during the dramatic rise in capital flows after the early 2000s. Our findings also indicate that capital flows influence the capital structures of different types of firms differently; increasing debt usage most in large private non-manufacturing firms.

On the whole, our empirical findings suggest that the trade-off theory is a better description of the capital structures of Turkish non-financial firms than the pecking-order theory. This contrasts sharply with the judgments of previous researchers on the Turkish economy, who typically take the opposite view. There are two main reasons behind this difference in judgments. First, our dataset is much larger (both within and across time) than the datasets used in past studies, which allows us to estimate the leverage effects of different leverage determinants much more precisely. Second, our dataset is much more comprehensive in terms of both firm types and leverage determinants, which enables us to do a much more comprehensive comparative test of the two theories. It is important to note the fact that our analysis of the relative economic significance of leverage determinants also strengthens the case for the trade-off theory.

Our findings also suggest where each of the two capital structure theories may be most fruitfully applied. In particular, we provide evidence that the trade-off theory is particularly successful in accounting for the capital structures of large private firms in the non-manufacturing sector. Moreover, the predictive power of the trade-off theory appears to be higher when the economic environment is relatively stable. Pecking order theory, on the other hand, although in our evaluation still not as powerful as the trade-off theory, appears to perform best for small public firms in the manufacturing sector and when the economic environment is relatively unstable.

The remainder of the paper proceeds as follows. Section 2 presents the links to the relevant theoretical and empirical literature on capital structure. Section 3 summarizes the hypotheses regarding the relationship between leverage and its determinants. Section 4 describes the data. Section 5 presents the main results and a number of robustness checks. Section 6 sheds light on the capital structure differences between firms of various types. Section 7 evaluates the predictive abilities of trade-off and pecking order theories in light of our empirical findings. Section 8 provides concluding remarks.

2 Main connections with the literature

The theory of corporate finance in a modern sense begins with the Modigliani and Miller (1958) theorem, which states that in a frictionless capital market a firm's market value is independent of how it finances its operations. Subsequent research emphasized important departures from the Modigliani-Miller assumption of frictionless capital markets such taxes, transaction costs, agency costs, adverse selection, lack of separability between financing and operations, time varying financial market opportunities, and investorclientele effects (Frank and Goyal 2008). These efforts led to the development of various capital structure theories, the two major theories being the trade-off and pecking order theories. Kraus and Litzenberger (1973) provides a classic statement of the theory that optimal capital structure reflects a single-period trade-off between the tax benefits of debt financing and the deadweight costs of potential bankruptcy. This standard version of the trade-off theory is referred to as the static trade-off theory.¹ The standard statement of the pecking order theory, on the other hand, is provided by Myers (1984) and Myers and Majluf (1984), according to which, firms follow a financing hierarchy in order to minimize the problem of asymmetric information between the firm's insiders and the outsiders.

Thanks to the trade off, pecking order, and other theories, we now know a lot more about the factors that determine firms' actual capital structure decisions. In particular, these theories have been helpful in identifying a large number of factors that seem to be correlated with firms' debt ratios. Frank and Goyal (2009) lists as many as 36 variables that are correlated with the leverage decisions of U.S. firms (see, also, Harris and Raviv 1991). Empirical studies typically employ a variety of variables from this list that can be justified using any of the existing theoretical models.

Most of the earlier empirical research attempts to test various theories using the developed countries, mainly the United States, as a laboratory (e.g. Bradley et al. 1984; Taggart 1985; Pozdena 1987; Titman and

Wessels 1988). In an influential study, Rajan and Zingales (1995) show that factors such as growth, size, profitability, and tangibility, which are correlated in the cross-section with firm leverage in the United States are similarly correlated in the other G-7 countries as well. Following Rajan and Zingales (1995), there has been an outpouring of empirical research on international comparisons of capital structure including, among others, Demirgüç-Kunt and Maksimovic (1996, 1999), Wald (1999), De Jong et al. (2008), and Antoniou et al. (2008). A particularly important study in this line of research has been Booth et al. (2001), which provides evidence that firms' capital structure decisions in developing countries are affected by the same variables as in developed countries. Since at least the mid-1990s, there are also a growing number of studies that explore the determinants of capital structure in individual developing countries.²

Our paper is most closely related to this last strand of literature since we analyze the capital structure of non-financial firms in a major developing economy, namely, Turkey.³ This literature studies the capital structure choice of manufacturing firms listed on Borsa Istanbul (BIST) (as the Istanbul Stock Exchange is called since 2013) from various angles and for different time periods. There are three exceptions: Demirhan (2009) focuses on non-manufacturing firms on the ISE, Okuyan and Taşçı (2010) use a dataset compiled by Istanbul Chamber of Industry that provides information on the largest 1,000 manufacturing firms in Turkey, and Aydın et al. (2006) use the same dataset as ours but restrict themselves only to a descriptive analysis of firms' capital structures. Cebenoyan et al. (1995) and Dincergök and Yalçıner (2011), on the other hand, carry out comparative studies of Turkey, Greece, Canada, and the U.S., and of Turkey, Brazil, Argentina, and Indonesia, respectively, again using data on listed manufacturing firms.

¹ Fischer et al. (1989) and Hennessy and Whited (2005) present different formulations of what is known as "dynamic trade-off theory", where the firm has a target level of leverage and deviations from that target are gradually removed over time. In this paper, trade-off theory refers to static trade-off theory, unless stated otherwise.

² See, for example, Wiwattanakantang (1999) for Thailand, Pandey (2004) for Malaysia, Huang and Song (2006) for China, Correa et al. (2007) for Brazil, Qureshi (2009) for Pakistan, and Espinosa et al. (2012) for Chile.

³ Previous studies on the Turkish economy include, among others, Durukan (1997, 1998), Gönenç (2003), Acaravcı and Doğukanlı (2004), Aydın et al. (2006), Sayilgan et al. (2006), Korkmaz et al. (2007), Yıldız et al. (2009), Demirhan (2009), and Okuyan and Taşçı (2010).

In all empirical studies, including those on the Turkish economy, some of the findings turn out to be consistent with the trade-off theory and others with the pecking order theory.⁴ One of the theories is judged to have greater explanatory power if more of the observed relationships are consistent with its predictions. What is clear, however, is that neither theory is uniformly better than the other theory. Therefore, our goal in this paper is to investigate the circumstances under which a given capital structure theory performs well, rather than look for "the theory" that captures reality in all possible circumstances. To achieve this goal, one must use a dataset that is comprehensive in terms of both firm types and leverage determinants suggested by theory. The dataset must also have a sufficiently long time span that will allow the researcher to analyze the effects of variations in the stability of the economic environment in which the firms operate. This is what we do in the present paper.

3 Hypothesis development

In this section, we develop the hypotheses about the nature of the relationships between leverage and various determinants through the lenses of the trade off and pecking order theories. Our discussion benefits, among others, from the reviews by Frank and Goyal (2008) and Harris and Raviv (1991). We first introduce our measures of leverage and then consider, in turn, firm-specific, tax-related, industry-specific, and macroeconomic determinants of leverage.

3.1 Measures of leverage

We consider three different measures of leverage: Short-term, long-term, and total debt over total assets. According to Rajan and Zingales (1995), this is a more appropriate measure of financial leverage than the ratio of liabilities (both short term and long term) to total assets because it provides a better indication of whether the firm is at risk of default any time soon and also a more accurate picture of past financing choices. Debt is classified as long-term if it has a maturity of at least 1 year and short-term otherwise.

Even though a complete analysis of debt maturity is beyond the scope of this paper,⁵ it is still important for us to distinguish between short-term and long-term leverage for two reasons. First, the trade-off and pecking order theories sometimes have different implications for the different types of debt. Second, short-term debt carries a number of risks for the financial and economic health of a company. The main financial risk is the so called maturity risk (the potential lack of liquid assets at the time of repayment of the loan) which exposes the borrowing firm to potential rollover difficulties and interest rate fluctuations.⁶ Moreover, the financial health of the corporate sector has serious implications for the health of the financial sector as well (Basel Committee on Banking Supervision 2011). On the other hand, when external finance is short term, it becomes more difficult for firms to improve their economic health by engaging in efficiency and capacity-enhancing investments. The fact that the majority of firms' debt is short-term in developing economies (Demirgüç-Kunt and Maksimovic 1999) coupled with the weaknesses in institutional and financial infrastructures and the volatile nature of the economic environment imply that these risks are more pertinent for developing economies than more stable advanced economies. In fact, there is ample evidence that short term debt did play an important role in the crises of Mexico 1994-1995, East Asia 1997-1998, and Brazil 1998-1999 (Schmukler and Vesperoni 2006).

⁴ For developed economies, support for the trade-off theory can be found, among others, in Antoniou et al. (2008) for for UK, US, Germany, France, and Japan, and Frank and Goyal (2009) for the US, whereas support for the pecking order theory can be found in Zoppa and McMahon (2002) for Australia. For developing economies, support for the trade-off theory can be found in Wiwattanakantang (1999) for Thailand, Huang and Song (2006) for China, and Espinosa et al. (2012) for Chile, whereas support for the pecking order theory can be found in Pandey (2004) for Malaysia, Korkmaz et al. (2007) for Turkey, Correa et al. (2007) for Brazil, and Qureshi (2009) for Pakistan.

⁵ There is a related but distinct strand of literature devoted to analyzing the determinants of corporate debt maturity. Barclay and Smith (1995) and Stohs and Mauer (1996) are notable early papers on the subject.

⁶ Another important risk is the exchange rate mismatch, which may occur due to a rise in foreign currency debt, while the firm's income is in domestic currency.

3.2 Determinants of leverage

3.2.1 Firm-specific determinants

Rajan and Zingales (1995) use four firm-specific independent variables in their study of capital structures: size, profitability, growth, and tangibility. Booth et al. (2001) add business risk to this list. We include all five firm-specific variables in our analyses.

3.2.1.1 Size Trade-off theory predicts a positive relationship between firm size and leverage. This is because larger firms are more diversified and have lower default risk. The pecking order theory, on the other hand, is generally interpreted as predicting a negative relationship, since large firms face lower adverse selection and can more easily issue equity compared to small firms. An overwhelming majority of empirical studies finds a positive relation between leverage and size. Following Titman and Wessels (1988) and Rajan and Zingales (1995), among others, we define size as the natural logarithm of total sales, adjusted for inflation.⁷

3.2.1.2 Profitability Trade-off theory is generally interpreted as predicting a positive relation between firm profitability and leverage. This is because default risk is lower and interest tax shields of debt are more valuable for profitable firms. Pecking order theory, on the other hand, predicts a negative relation between leverage and profitability, as profitable firms can use earnings to fund investment opportunities and hence have less need for external debt. Empirical tests find the relation to be robustly negative. Following Titman and Wessels (1988) and De Jong et al. (2008), among others, we define profitability as operating income over total assets.

3.2.1.3 Tangibility We use tangibility as a proxy for the type of assets. The trade-off theory predicts a positive relation between leverage and tangibility. This is because tangible assets are easier to collateralize and they suffer a smaller loss of value

when firms go into distress. In addition, since firms tend to match the maturity of assets with that of liabilities (Stohs and Mauer 1996), tangibility should be positively correlated with long-term leverage. The pecking order theory, on the other hand, is generally interpreted as predicting a negative relation between leverage and tangibility, since the low information asymmetry associated with tangible assets makes the issuance of equity less costly (Harris and Raviv 1991). Empirical studies generally find a positive correlation between tangibility and total and long-term leverage. Following Rajan and Zingales (1995) and Demirgüç-Kunt and Maksimovic (1999), we define tangibility as the ratio of net fixed assets to total assets.

3.2.1.4 Growth Trade-off theory predicts a negative relation between leverage and firm growth. Intangibility of the assets of growth firms implies that they lose more of their value in the event of financial distress. By contrast, the pecking order theory predicts a positive relation between leverage and growth. This is because internal funds are unlikely to be sufficient to support investment opportunities for high growth firms, which increases their demand for external debt. Although the results are mixed, most empirical work finds the relation between leverage and growth to be negative. Since our sample consists of both private and public firms, we cannot use market measures such as market-to-book ratios to proxy for growth. Our proxy is the percent change in sales as in Frank and Goyal (2009) and Schoubben and Van Hulle (2004).

3.2.1.5 Business risk Both the trade-off and pecking order theories predict a negative relationship between leverage and business risk. The trade-off theory implies that the expected cost of financial distress increases with risk, at least for reasonable parameter values. In addition, the probability of wasting interest tax shields increases when earnings are less than tax shields (Frank and Goyal 2008). Both forces work to reduce leverage. From a pecking order perspective, business risk exacerbates the adverse selection between firms and creditors. Most empirical work finds a negative relation between leverage and risk. In this study, business risk is defined as the standard deviation of operating income over total assets over the past 3 years (including the current year) as in De Jong et al. (2008).

⁷ We use total sales rather than total assets to alleviate the problem of multicollinearity since many of our variables are scaled by total assets, including those for debt ratios. These two measures are highly correlated, indicating that each of them should be a sound proxy for size.

3.2.2 Tax-related determinants

The two tax-related determinants we consider are corporate taxes and non-debt tax shields.⁸ These determinants naturally fit in with the trade-off theory.

3.2.2.1 Corporate taxes Trade-off theory predicts a positive relationship between corporate tax rates and leverage. This is because features of the tax code allow interest payments to be deducted from the tax bill but not dividend payments, which provides a tax advantage for debt. The effect of taxes on debt ratios, however, has been difficult to clearly identify in the data and the available evidence is rather mixed (see, for example, Frank and Goyal 2008; Antoniou et al. 2008). One explanation for this might be the uncertainty about what would constitute a good proxy for tax effects. Another explanation is that transaction costs make it difficult to identify tax effects even when they are an element of the firm's problem (Hennessy and Whited 2005).

3.2.2.2 Non-debt tax shields DeAngelo and Masulis (1980) were probably the first to formally introduce the concept of non-debt tax shields to the literature. Examples of such shields include depreciation deductions, depletion allowances, and investment tax credits. These shields can be considered as substitutes for the corporate tax benefits of debt financing. Accordingly, firms with higher amounts of non-debt tax shields will choose to have lower levels of debt. Thus, the trade-off theory predicts a negative relationship between leverage and non-debt tax shields. More often than not, empirical studies find results that are supportive of this prediction.

Rather than including corporate taxes and non-debt tax shields separately in our analyses, we use a single indicator that simultaneously takes into account the presence of both effects. Whether a firm actually enjoys a positive tax advantage for debt financing depends on the trade-off between these two effects.

$$PDTS = \begin{cases} I_{it} + \frac{T_{it}}{\tau_t} & \text{if } T_{it} > 0\\ 0 & \text{if } T_{it} = 0 \end{cases},$$

where I_{it} and T_{it} denote, respectively, interest payments and tax payments by firm *i* at time *t* and τ_t denotes the statutory corporate tax rate at time t. PDTS is gross earnings minus non-debt tax shields and Shuetrim et al. (1993) show that it is equal to the sum of interest paid and taxable income after all non-debt tax deductions have been made as shown in the above expression.⁹ They scale this sum by a firm's total assets to get their final proxy of net tax shields. Note that when tax payments are zero (i.e. tax is exhausted), the relative proportions of income shielded by interest payments and by non-debt tax shields cannot be determined and hence PDTS is set to zero. In order to control for this possibility, we follow Shuetrim et al. (1993) and include firms' state of tax exhaustion as a separate regressor in our analyses. It is a dummy variable that is equal to 1 for all observations when the tax paid by a firm is equal to zero. Its value can be interpreted as the mean effect of PDTS on leverage taken over all observations with zero tax payments. From a trade-off perspective, the predicted relation between this dummy variable and leverage is positive. Note that both PDTS and tax exhaust determinants can be calculated at the firm level.

3.2.3 Industry-specific determinants

The extant literature reviewed in articles such as Harris and Raviv (1991) and Frank and Goyal (2008) suggests that industry membership may be an important determinant of firms' capital structures. According to Frank and Goyal (2009), this is mainly because industry reflects a number of otherwise omitted factors common to all firms. For example, supply and demand conditions or the extent of competition may differ from industry to industry. From a trade-off perspective, therefore, although imperfect, the *industry median*

⁸ Studies such as Miller (1977) and Rajan and Zingales (1995) recommend including both personal and corporate taxes in studies of capital structure. However, in Turkey, tax rates on equity and debt income at the personal level are extremely complicated and have gone through several reforms during the past two decades. This makes it almost impossible to come up with good indicators of personal tax rates on different sources of income that would also be consistent over time. As a result, we are forced to do away with personal taxes in our analyses.

⁹ This measure is also desirable because it takes into account the actual statutory tax rates and the fact that firms focus on the amount of income that can be shielded from tax using interest payments.

leverage is likely to be a proxy for firms' target capital structure. Not entirely coincidentally, empirical evidence on industry effects is rather mixed. For instance, while Hovakimian et al. (2001) find that firms adjust their debt ratios towards industry median debt ratios, Mackay and Phillips (2005) provide evidence that there is significantly more variation in leverage within industries than across industries. For a given year, we define industry median leverage as the median of (short, long, or total) debt to total assets in that industry (that is, one for each of the two broad industry categories, manufacturing and non-manufacturing).

3.2.4 Macroeconomic determinants

Many studies including Deesomsak et al. (2004) and de Jong et al. (2008) show that the health and stability of the economic environment exert considerable effect on the firms' capital structures. In order to explore the influence of the economic environment on firms' capital structures, we include key macroeconomic variables in our analyses. Specifically, we include inflation and GDP growth as indicators of the general economic environment and the size of capital flows as an indicator of financial development.

3.2.4.1 Inflation According to Taggart (1985), features of the tax code in the U.S. increases the real value of interest tax deductions on debt when inflation is expected to be high. Thus, the trade-off theory predicts a positive relationship between leverage and expected inflation. By contrast, it is hard to see why inflation would matter for firms' leverage decisions in a model of pecking order (Frank and Goyal 2009). Empirical studies generally find a positive relation between leverage and inflation. In the absence of inflation expectations data that spans the whole sample period, we follow previous studies and use data on the realized inflation. Specifically, we use the percentage change in the annual consumer price index (CPI) as a rough proxy for expected inflation. It is important to note that the debt-bias in the U.S. tax system alluded to by Taggart (1985) is also a feature of the tax systems of many countries around the world, including that of Turkey.

3.2.4.2 GDP growth Real Gross Domestic Product (GDP) growth can be viewed as a measure of the growth opportunities available to firms in an economy. In a high growth environment, the scarcity of firms' tangible

 Table 1
 Definitions and hypotheses

	Definition	Pecking order theory	Trade- off theory
Debt ratios			
Short-term leverage	Short-term debt/total assets	N/A	N/A
Long-term leverage	Long-term debt/total assets	N/A	N/A
Total leverage	Total debt/total assets	N/A	N/A
Firm-specific	determinants		
Size	Natural logarithm of total net sales	_	+
Profitability	Operating income/total assets	_	+
Tangibility	Net fixed assets/total assets	_	+
Growth	Percent change in net sales	+	_
Business risk	SD of operating income/	-	-

Tax-related determinants

Potential debt tax shields	Interest payments + tax payments/corporate tax rate	?	+
Tax exhaust	Dummy equal to 1 if firm pays no tax	?	+
Industry-spec	ific determinants		
Industry median leverage	Industry median of (short, long, or total) debt to total assets	?	+

Macroeconomic determinants

Inflation	Percent change in CPI	?	+
GDP growth	Percent change in real GDP	+	-
Capital flows	Net capital flows/GDP	?	?

This table describes the dependent and independent variables and the theoretical predictions for the relations between the dependent and independent variables. While the sign "+" denotes a positive association with leverage, the sign "-" denotes a negative association

assets relative to available investment opportunities implies a higher loss of value when firms go into distress. Hence, the trade-off theory predicts a negative relation between leverage and GDP growth. By contrast, the pecking order theory predicts a positive relation between leverage and macroeconomic growth, since a high ratio of growth opportunities to internal funds would imply a greater need for external finance. Empirical studies generally find a negative association between leverage and macroeconomic growth (see, for example, Demirgüç-Kunt and Maksimovic 1996). Following common practice, we define GDP growth as the percent change in the annual real GDP.

3.2.4.3 Capital flows Empirical studies such as Demirgüç-Kunt and Maksimovic (1996) and Antoniou et al. (2008), among others, provide evidence on the importance of capital markets for firms' capital structures. The size and structure of capital markets play a key role in determining the availability and allocation of funds to various types of firms within an economy. Domestic capital markets, in turn, are heavily shaped by the flows of international capital in many developing economies, including that of Turkey (see, for example, Kose et al. 2009). While inflows of capital lead to increases in the size of domestic capital markets, outflows lead to declines. In order to study the impact of capital flows (and hence financial development) on firms' capital structures, we include the ratio of net capital flows to GDP as an additional explanatory variable in our regression equations.

Table 1 summarizes the definitions of various leverage measures and leverage determinants as well as the theoretical predictions for the relations between the dependent and independent variables.

4 Data

Our firm- and industry-specific data come from the survey-based Sectoral Balance Sheets (SBS) dataset of the CBRT, which is the largest source of annual balance sheet and income statement data on Turkish non-financial firms.^{10,11} Our sample period covers the years 1996–2009.

In our sample, there are on average about 9000 firms each year, of which roughly 2 percent are publicly traded firms. The sample includes nearly all large firms in Turkey as well as a large number of small and medium-sized enterprises (SMEs) and micro-sized firms.¹² Altogether our sample firms employ a total of about 1.5 million workers each year, accounting for roughly 11 % of employment in Turkey during 1996–2009.¹³ This is substantially larger than the corresponding figure for non-financial firms listed on BIST during the same period (225 thousand workers), indicating that the SBS dataset is significantly more representative of the population of Turkish non-financial firms than BIST.

The two broad industry categories we use are 'manufacturing' and 'non-manufacturing'. The manufacturing industry consists of 13 sub-industries (comprising roughly 53 % of all firms) whereas the non-manufacturing industry consists of 4 sub-industries (comprising roughly 47 % of all firms).¹⁴ Our

¹³ The ratio of the total number of workers employed by the SBS firms to non-financial sector employment in Turkey is likely to be considerably greater than 11 %. Unfortunately, the non-financial sector employment number is not published by the Turkish Statistical Institute. Moreover, although the lack of appropriate data makes the calculation impossible, the fact that almost all large firms and a large number of medium-sized firms are included in our dataset implies that our sample is likely considerably more representative of the Turkish non-financial sector in terms of sales; at least about 40–50 % in any given year.

¹⁴ The manufacturing industries are: (1) food products, beverages, and tobacco, (2) textiles and textile products, (3) leather and leather products, (4) wood and wood products, (5) pulp, paper, paper products, publishing, and printing, (6) chemicals, chemical products, and man-made fibers, (7) rubber and plastic products, (8) other non-metalic mineral products, (9) basic metals and fabricated metal products, (10) machinery and equipment N.E.C., (11) electrical and optical equipment, (12) transport equipment, and (13) furniture, manufacturing N.E.C. The non-manufacturing industries are: (1) construction, (2) wholesale and retail trade, (3) hotels and restaurants, and (4) transport, storage, and communications. On the other hand, we exclude some industries from our analyses. These are: (1)

¹⁰ For confidentiality purposes, the SBS data are not available to the general public in its raw form. However, an aggregated version of the data is available on the Bank's website at http:// www.tcmb.gov.tr/sektor/mainmenu.htm along with a report based on the data annually.

¹¹ SBS firms report unconsolidated balance sheets. As noted by Rajan and Zingales (1995), this may cause firms to incorrectly appear to have lower leverage than otherwise identical firms

Footnote 11 continued

who report consolidated balance sheets. This should not be a serious problem for our analyses since most of the sample firms are stand-alone enterprises. Still, some care is warranted in interpreting the results.

 $^{^{12}}$ By micro-sized, small and medium-sized, and large-sized firms we mean firms with 1–9 employees, 10–249 employees, and >250 employees, respectively. According to this classification, there are on average about 1,350 (15 %) micro-sized firms, 6,390 (71 %) SMEs, and 1,260 (14 %) large firms in our sample. Classifications based on sales or assets would likely yield similar size distributions.

Table 2Descriptivestatistics

	Observations	First quartile	Mean	Median	Third quartile	Coefficient of variation
Debt ratios						
Short-term leverage	74,155	3.48	20.37	15.08	31.57	97.12
Long-term leverage	74,155	0.00	6.81	0.00	7.73	195.37
Total leverage	74,155	7.73	27.20	23.87	42.17	81.18
Firm-specific determina	nts					
Size	74,155	12.93	14.01	14.04	15.15	12.72
Profitability	74,155	0.01	0.09	0.07	0.15	156.83
Tangibility	74,155	10.82	31.36	25.87	46.78	78.39
Growth	74,155	-18.94	44.35	1.57	24.07	1,005.70
Business risk	74,155	0.03	0.07	0.05	0.10	100.96
Tax-related determinant	ts					
Potential debt tax shields	74,155	0.00	0.04	0.00	0.04	188.91
Tax exhaust	74,155	0.00	0.37	0.00	1.00	130.65
Industry-specific determ	inants					
Short-term industry median leverage	28	10.68	13.64	12.69	17.28	30.34
Long-term industry median leverage	28	0.00	0.46	0.00	0.04	217.86
Total industry median leverage	28	17.05	22.04	22.78	26.87	28.32
Macroeconomic determ	inants					
Inflation	14	10.51	50.61	54.40	84.64	70.14
GDP growth	14	0.66	3.56	6.16	7.58	143.92
Capital flows	14	2.14	2.91	3.02	4.85	104.37

This table presents the descriptive statistics for the dependent and independent variables described in Table 1. All variables (other than industry-specific and macroeconomic variables) are winsorized at 0.50 % level in both tails of the distribution. The sample period is 1996-2009. All numbers are reported in two decimal places. Agricultural firms, financial firms and firms that operate in regulated industries are excluded

data on macroeconomic and tax-related variables, on the other hand, are collected from a variety of sources including SBS, BIST, Turkish Statistical Institute, Undersecretariat of Treasury of the Republic of Turkey, and World Development Indicators. To minimize the effects of outliers in the data on our results, we use winsorization, in which the most extreme tails of the distribution are replaced by the most extreme value that has not been removed (Frank and Goyal 2008). Following common practice, we winsorize each tail at 0.5 percent. The final sample is an unbalanced panel of 11,726 firms with 74,155 firmyear observations. We do not have 14 years of data for all firms because each year some firms enter or exit the sample.¹⁵

Footnote 14 continued

agriculture, hunting, and forestry, (2) fishing, (3) mining, (4) electricity, gas, and water supply, (5) real estate, renting, and business activities, (6) education, (7) health and social work, and (8) other community, social, and personal service activities. Unlike the industries included in our analyses, these industries are generally under the influence of various sorts of government intervention that distort the operation of market forces.

¹⁵ Possible bias introduced by firm entry or exit is discussed in Sect. 5.4.

Table 3 Determinants of leverage: full sample results

	Short-term Leverage	Long-term leverage	Total leverage
Firm-specific determinants			
Size	1.079***	0.843***	1.918***
	(0.100)	(0.0734)	(0.114)
Profitability	-15.260***	-5. 358***	-20.690***
	(0.677)	(0.454)	(0.755)
Tangibility	-0.072^{***}	0.097***	0.025***
	(0.006)	(0.005)	(0.007)
Growth	-0.0002	0.0000	-0.0002
	(0.0001)	(0.0001)	(0.0002)
Business risk	-1.527	-4.245***	-5.692***
	(1.189)	(0.751)	(1.306)
Tax-related determinants			
Potential debt tax shields	40.080***	10.350***	50.280***
	(1.461)	(0.938)	(1.566)
Tax exhaust	4.221***	2.261***	6.500***
	(0.170)	(0.122)	(0.189)
Industry-specific determinants			
Industry median leverage	0.329***	0.423***	0.234***
	(0.041)	(0.057)	(0.021)
Macroeconomic determinants			
Inflation	0.037***	0.015***	0.057***
	(0.003)	(0.002)	(0.004)
GDP growth	-0.000	-0.105***	-0.071***
	(0.016)	(0.013)	(0.019)
Capital flows	0.106***	0.119***	0.188***
	(0.0328)	(0.0226)	(0.0361)
Number of observations	74,155	74,155	74,155
Number of firms	11,726	11,726	11,726
Adjusted R-squared	0.650	0.612	0.673

This table presents the results from the estimation of our fixed effects panel regression Eq. (1): $L_{it} = \beta_0 + \sum_k \beta_k F_{k,it} + \mu_i + \varepsilon_{it}$, where L_i is one of the three measures of leverage (short, long, or total leverage) of firm *i* in year *t*; *F* is the vector of the four types of leverage determinants (firm-specific, tax-related, industry-specific, and macroeconomic determinants); μ_i are the time-invariant unobservable firm-specific effects; and ε_i is the error term. The variables are the same as those defined in Table 1 and all (except growth) are reported in three decimal places. Heteroskedasticity and serial correlation robust standard errors are reported in parentheses

***, **, and * Significance levels at the 1, 5, and 10 % levels, respectively

Table 2 presents the descriptive statistics for all of our variables during 1996–2009. The median is below the mean for all three debt ratios. The divergence between the mean and median debt ratios is larger for the long-term debt ratio as the majority of firms have little to no long-term debt. Moreover, firm growth rates have the largest variance, with the mean significantly greater than the median.

5 Empirical model and results

In this section, we present our empirical model, discuss the main results, and perform a number of robustness checks.

5.1 Empirical model

We model leverage as a function of various determinants discussed in the previous section. Specifically, we estimate the following fixed effects panel data model:

$$L_{\rm it} = \beta_0 + \sum_k \beta_k F_{k,\rm it} + \mu_i + \varepsilon_{\rm it},\tag{1}$$

where L_{it} is one of the three measures of leverage (short, long, or total leverage) of firm *i* in year *t*; *F* is the vector of the four types of leverage determinants (firm-specific, tax-related, industry-specific, and macroeconomic determinants); μ_i are the time-invariant unobservable firm-specific effects; and ε_{it} is the error term. We estimate Eq. (1) using standard errors that are robust to heteroskedasticity and serial correlation. We calculate robust standard errors using Newey and West (1987)'s variance estimator which produces consistent estimates when there is heteroskedasticity and autocorrelation in standard errors.

5.2 Results

Table 3 presents the results from estimating Eq. (1). In what follows, we summarize the results and evaluate them in light of theoretical hypotheses and previous empirical findings.

5.2.1 Firm-specific determinants

The first five rows of Table 3 display the estimated coefficients for our firm-specific determinants. The coefficients of size, profitability, and tangibility are significant at the 1 % level in all leverage equations. Size is positively associated with all three debt ratios, suggesting that *ceteris paribus* large firms have more debt in their capital structures. By contrast, the relation between profitability and various debt ratios is negative, implying that more profitable firms have lower debt ratios. While our result on size is consistent with the prediction of trade-off theory, our result on

profitability is consistent with the pecking order theory. These findings are also in line with the empirical evidence found in previous studies.

Tangibility is negatively associated with short-term leverage but positively associated with long-term and total leverage. Thus, firms with more tangible assets tend to have more long-term and less short-term debt in their capital structure. The fact that the signs of the estimated coefficients of the tangibility determinant are opposite in the short- and long-term leverage equations can be interpreted as evidence that firms in Turkey match the maturity of their assets with their liabilities. Demirgüç-Kunt and Maksimovic (1999) report similar findings in their sample of nineteen developed and developing countries (Turkey is included) and Gönenç (2003) for industrial firms listed on the ISE. Overall, the results are consistent with the prediction of trade-off theory but not with the pecking order theory.

Finally, while firm growth appears to be unrelated with leverage, the estimated coefficient of our business risk variable is significantly negative in the long-term and total leverage equations. So, increases in a firm's riskiness reduce the level of long-term debt in its capital structure but does not have a significant effect on the level of short-term debt relative to total assets. This is consistent with the view that firms that are viewed as risky by creditors find it more difficult to borrow long-term (see, for example, Diamond 1991 or Demirgüç-Kunt and Maksimovic 1999). Our results concerning the business risk determinant can be understood within the framework of either trade-off or pecking order theories.

5.2.2 Tax-related determinants

The upper middle section of Table 3 presents the estimated coefficients for our *PDTS* and tax exhaustion variables. The coefficients of both variables are significant at the 1 percent level in all leverage equations and have the signs predicted by the trade-off theory. The positive and significant coefficient on PDTS suggests that the tax advantage of debt is greater than the tax advantage due to non-debt shields. The positive and significant coefficient on the tax exhaustion dummy, on the other hand, indicates that the tax distortions caused by the tax system are also important for firms that pay no tax.

5.2.3 Industry-specific determinants

The lower middle section of Table 3 presents the estimated coefficients for our industry median debt ratios. The coefficient of industry median leverage is positive and highly significant in all three leverage equations. Moreover, the association appears to be quantitatively strong: A 10 percentage point increase in the industry median short-term, long-term, and total leverage increases the average firm's short-term, long-term, and total leverage points, respectively. From a trade-off perspective, these findings can be interpreted as evidence of target adjustment behavior in leverage.

5.2.4 Macroeconomic determinants

The bottom section of Table 3 presents the estimated coefficients for our macroeconomic determinants. In all equations, inflation is positively related with leverage and with coefficients that are significant at 1 percent. Therefore, firms' indebtedness increases with inflation. This finding is consistent with the tradeoff theory in which, given the tax-deductibility of nominal interest payments, an inflation-induced increase in nominal interest rates increases the tax advantage of debt financing. In addition, the impact of inflation appears to be quite strong: The fact that inflation has come down from over 80 percent to less than 10 percent between 1996 and 2009 would suggest roughly a 5 percent decline in the average firm's total leverage due solely to the fall in inflation, all else equal.

On the other hand, the coefficient on GDP growth is negative and highly significant in the long-term and total leverage equations, but not the short-term leverage equation. One explanation of this finding might be that the scarcity of firms' tangible assets relative to available investment opportunities is exacerbated in a high growth environment. The implied higher loss of value in the event of a distress, in turn, reduces firms' ability to raise longer-term debt. Put differently, creditors reduce the amount of long-term debt they extend because it allows them to review the firms' decisions more frequently and, if necessary, to vary the terms of financing before sufficient losses have accumulated to make default by the borrower optimal (Diamond 1991 and Demirgüç-Kunt and Maksimovic 1999). Overall, our findings on inflation and GDP growth appear to be consistent with the trade-off theory.

Finally, capital flows are positively and highly significantly correlated with all three measures of leverage, indicating that capital flows do in fact play an important role in shaping firms' capital structures. Consistent with expectations, measures of leverage rise with inflows of capital and fall with outflows. While capital inflows are in principle beneficial for developing economies, particularly for those with low saving rates such as Turkey,¹⁶ they can also lead to economic overheating, excessive currency appreciation, or pressures in particular sectors of the economy, such as sectoral credit booms or asset price bubbles (Ostry et al. 2011). In particular, rapid growth in private indebtedness induced by capital inflows might be characteristic of the early stages of financial instability (Mishkin 1997) and might cause banking crises (Reinhart and Rogoff 2011). Our results, therefore, indicate that the surge in international capital flows since the early 2000s, which were only briefly interrupted by the global financial crisis, continue to pose serious risks to economic and financial stability. Reflecting these concerns, policymakers in many developing economies have been very keen on finding appropriate ways to deal with the adverse effects of capital flows, particularly in the aftermath of the crisis. For example, Brazil and South Korea have chosen to implement capital control measures whereas Turkey and Indonesia have preferred macroprudential measures.

5.3 Economic significance of leverage determinants

In the previous subsection, we investigated whether a given leverage determinant, holding other determinants constant, was statistically significant in accounting for various debt ratios. However, statistical significance need not imply economic significance. In order to study the relative economic significance of various leverage determinants, we standardize all

¹⁶ Dell'Ariccia et al. (2008)among others, argue that capital inflows can ease financing constraints for productive investment projects, foster the diversification of investment risk, promote intertemporal trade, and contribute to the development of financial markets.

 Table 4
 Economic significance of leverage determinants

	Standardized	Standardized coefficients				
	Short-term leverage	Long-term leverage	Total leverage	Short-term leverage	Long-term leverage	Total leverage
Firm-specific determinants						
Size	0.097	0.113	0.155	4	2	2
Profitability	-0.106	-0.055	-0.129	2	4	4
Tangibility	-0.089	0.179	0.028	5	1	7
Growth	-0.005	0.001	-0.004	_	_	_
Business risk	-0.006	-0.024	-0.019	_	9	9
Tax-related determinants						
Potential debt tax shields	0.142	0.054	0.160	1	5	1
Tax exhaust	0.103	0.082	0.142	3	3	3
Industry-specific determinan	ts					
Industry median leverage	0.063	0.037	0.059	6	7	6
Macroeconomic determinant	ts					
Inflation	0.056	0.033	0.077	7	8	5
GDP growth	0.000	-0.038	-0.016	_	6	10
Capital flows	0.014	0.024	0.023	8	10	8

This table presents the findings on the relative economic significance of various leverage determinants. The estimated standardized coefficients show the impact of a one standard deviation change in an independent variable on the dependent variable. The magnitudes of the standardized coefficients are comparable across different leverage determinants. The left panel shows these results. The right panel sorts the leverage determinants according to their economic significance, where the ordering is based on the absolute value of the standardized coefficients. When a variable has a statistically insignificant coefficient it is denoted by "–" and left out of the economic significance rankings. The variables are the same as those defined in Table 1 and all are reported in three decimal places. Heteroskedasticity and serial correlation robust standard errors are reported in parentheses

variables and rerun our regressions. Now, the estimated standardized coefficients show the impact of a one standard deviation change in an independent variable on the dependent variable. Accordingly, the magnitudes of the standardized coefficients are comparable across different leverage determinants for a given leverage equation (short-term, long-term, or total leverage). The left panel of Table 4 reports these results. The right panel of Table 4 sorts the leverage determinants according to their economic significance, where the ordering is based on the absolute value of the standardized coefficients.

The results indicate that PDTS, our tax-shield measure, is the most economically significant determinant for firms' total leverage, with size a very close second. A one standard deviation increase in PDTS increases total leverage by 0.1595 standard deviations. PDTS is also the most important determinant for short-term leverage, followed by profitability. For long-term leverage, tangibility is by far the most economically important determinant, followed by size. Tax exhaust is consistently among the top three important determinants.

A few general lessons can be drawn from these results. Tax-related determinants are overall the most economically significant leverage determinants, suggesting that they in fact do play an important role in shaping firms' capital structures. This finding is particularly interesting in light of the fact that tax effects on leverage have historically been notoriously difficult to detect in the data. A close second in terms of economic significance is firm-specific determinants such as size, tangibility, and profitability, while business risk and firm growth play little or no role. Industry median leverage, on the other hand, appears to have a fairly important influence on firms' leverage decisions. Macro-level determinants generally score lower than firm-level (i.e. tax-related and firm-specific) determinants and industry-level determinants in the economic significance rankings, which is perhaps not surprising given that they have only a single observation per year. Among the macro-level determinants, inflation stands out both in terms of statistical and economic significance, even surpassing in ranking the industry median debt ratio in the total leverage equation. The only case in which inflation is not the highest ranking macro-level determinant is the longterm leverage equation, where it is second only to GDP growth and only by a small margin. These findings are particularly important in light of the fact that inflation in Turkey has only recently come down to single digits after decades of extremely high levels.

A second general lesson is that the economic importance of a given leverage determinant is different for different measures of leverage. For example, while tangibility appears to be a key determinant of long-term leverage, it is not as important for short-term leverage. A third and final lesson is that leverage determinants that are more closely associated with the trade-off theory such as tax-related determinants and size generally have higher rankings (in addition to having the correct signs and higher levels of statistical significance as shown in Sect. 5.2) than those that are more closely associated with the pecking order theory such as profitability. The evidence in favor of the trade-off theory is particularly strong for the long-term and total debt ratios.

5.4 Robustness checks

We perform a number of checks to confirm that our results are robust.¹⁷ First, it doesn't matter whether the data are winsorized or not; estimated coefficients are qualitatively and quantitatively very similar. This suggests that outliers in our dataset are not a significant problem for estimates.

Second, the fact that some firms in our dataset enter or exit the sample might potentially induce attrition bias in our estimations. To see if this is the case, we also estimate our empirical model only for those firms that have data for *T* years or more, where T = 2, 3,...,14 (T = 1 corresponds to the full sample results we report in Sect. 5.2). We find no evidence of the impact of attrition bias on our results.

Third, all the variables (i.e. dependent as well as independent) in our estimations are measured in year t, as is typically done in the capital structure literature. Nevertheless, in order to take the possibility that year t - 1 determinants are more relevant than year t determinants for a firm's capital structure in year t, we re-estimate our equations this time replacing the current values of our independent variables with the lagged values. According to the results we do not report here, the results are qualitatively virtually the same as before, suggesting that from a capital structure perspective the information contents of year t and year t - 1 independent variables are qualitatively quite similar.

Fourth, apart from the net capital inflow, the supply of domestic capital can in principle also affect firms' capital structure decisions. In order to explore this possibility, we also included as independent variables the size of debt markets (domestic credit to the private sector over GDP), the size of equity markets (stock market capitalization over GDP), and the size of government debt (total government debt over total domestic non-financial debt) in our regression equations. We find that the size of debt markets is positively correlated with long term leverage (at 5 percent significance level) and negatively correlated with short-term leverage (at 10 percent significance level). So, the deepening of debt markets appears to reduce firms' short-term leverage while increasing longterm leverage. This finding is not surprising given the fact that banks have a comparative advantage in the provision of long-term debt. Second, unsurprisingly, we find that the size of equity markets is negatively correlated with all measures of leverage and its coefficients are highly significant (at the 1 percent level). Finally, we find that the size of government borrowing is unrelated with firms' leverage decisions, suggesting that government borrowing does not crowd out private borrowing during our sample period. Unreported results also indicate that the relationships between leverage and the rest of the independent variables are both quantitatively and qualitatively very similar to the previous case.

Finally, we split the sample into two seven-year periods, namely, 1996–2002 and 2003–2009, to test for any structural breaks in the firms' capital structure

¹⁷ To economize on space, we do not report detailed results for our robustness analyses, except for the last one. However, these results are available upon request.

	1996–2002	1996–2002			2003–2009		
	Short-term leverage	Long-term leverage	Total leverage	Short-term leverage	Long-term leverage	Total leverage	
Firm-specific determin	nants						
Size	0.074	0.442***	0.519**	1.185***	0.721***	1.890***	
	(0.212)	(0.121)	(0.222)	(0.121)	(0.095)	(0.141)	
Profitability	-14.100***	-3.970***	-18.010 ***	-16.570***	-6.660***	-23.320***	
	(0.986)	(0.583)	(1.044)	(1.001)	(0.740)	(1.141)	
Tangibility	-0.036***	0.070***	0.035***	-0.060***	0.124***	0.064***	
	(0.0106)	(0.00748)	(0.0113)	(0.00825)	(0.00663)	(0.00918)	
Growth	0.0002	0.0002	0.0004	-0.0003**	0.0001	-0.0003	
	(0.0005)	(0.0003)	(0.0004)	(0.0002)	(0.0001)	(0.0002)	
Business risk	-3.927**	-2.838***	-6.578***	-2.674	-3.433***	-6.165***	
	(1.778)c	(0.965)	(1.864)	(1.733)	(1.200)	(1.931)	
Tax-related determina	ints						
Potential debt tax	31.890***	4.263***	35.970***	54.620***	16.120***	70.820***	
shields	(1.756)	(1.039)	(1.807)	(3.516)	(2.481)	(3.701)	
Tax exhaust	6.188***	2.205***	8.390***	2.968***	1.837***	4.828***	
	(0.331)	(0.218)	(0.349)	(0.208)	(0.159)	(0.230)	
Industry-specific deter	rminants						
Industry median	0.453***	a	0.443***	0.412***	0.135	0.206***	
leverage	(0.127)		(0.090)	(0.099)	(0.101)	(0.041)	
Macroeconomic deter	minants						
Inflation	0.027*	0.007	0.030**	0.060***	0.038***	0.132***	
	(0.015)	(0.005)	(0.013)	(0.017)	(0.012)	(0.016)	
GDP growth	0.033	-0.033*	0.022	-0.011	-0.231***	-0.166***	
	(0.028)	(0.018)	(0.028)	(0.024)	(0.030)	(0.039)	
Capital flows	0.039	-0.018	-0.042	0.101**	0.224***	0.311***	
	(0.074)	(0.038)	(0.076)	(0.045)	(0.027)	(0.046)	
Number of observations	31,732	31,732	3,1732	40,717	40,717	40,717	
Number of firms	7,311	7,311	7,311	8,447	8,447	8,447	
Adjusted R-squared	0.738	0.701	0.758	0.726	0.711	0.763	

Table 5 Robustness of full sample results to alternative time periods

This table presents the findings on the robustness of full sample results considering separately the period before 2002 and the period after 2002. The variables are the same as those defined in Table 1 and all (except growth) are reported in three decimal places. Heteroskedasticity and serial correlation robust standard errors are reported in parentheses

***, **, and * Significance levels at the 1, 5, and 10 % levels, respectively

^a Since most of the firms did not have long-term debt before 2002, the median debt for manufacturing and non-manufacturing industries were zero. Thus, there is no variation in industry median leverage variable in the 1996–2002 period and its coefficient cannot be estimated

choices. ¹⁸ The results displayed in Table 5 indicate that firm-specific and tax-related leverage determinants generally have the same pattern of signs and significance as those we obtained for the full sample. The only significant change here is that the coefficient of firm growth turns significant in the short-term leverage equation in the 2003-2009 subsample, perhaps reflecting the improved growth prospects during this period. It is also interesting to note that the relative magnitude of the coefficient on tax exhaust (PDTS) is larger in the former (latter) period, which suggests that the tax distortions caused by the tax system during the earlier period are larger (smaller) for firms that pay no tax (pay tax). Also, the coefficient of business risk in the short-term leverage equation turns significant in the 1996–2002 subsample, most likely a reflection of the volatile state of the economy during this period. Finally, there are some changes in the signs and significance of some of the industry-level and macro-level determinant coefficients.¹⁹ Specifically, the coefficient of industry median debt ratio turns insignificant in the long-term leverage equation in both subsamples and there is a loss of significance in the coefficients of inflation, GDP growth, and capital flow variables. In the case of capital flows, the contrast is particularly stark, where the coefficient is significant only in the 2003-2009 subsample, most likely reflecting the surge in capital inflows during this period. Overall, our results indicate that the predictive power of the trade-off theory is higher in the 2003–2009 period, during which the economic environment in Turkey was substantially more stable.

6 Capital structures of firms in various circumstances

Although both the trade-off theory and the pecking order theory have their strengths and weaknesses, our analysis thus far indicates that the former provides a considerably better framework than the latter to understand the capital structure of the average Turkish non-financial firm. However, as Myers (2003) argued, different factors might affect different types of firms in fundamentally different ways. To see if this is the case, we now systematically investigate the capital structure differences of three main types of firms: manufacturing versus non-manufacturing firms, large versus small firms, and public versus private firms. These exercises can also be viewed as additional robustness checks on our main results in Sect. 5.

Figure 1 displays the ratios of short-term, longterm, and total indebtedness by industry membership, firm size, and stock market listing averaged over the whole sample period. Panel A of Fig. 1 shows that while manufacturing firms generally have higher debt ratios than non-manufacturing firms, the difference is more apparent in the short-term debt ratios. In Panel B of Fig. 1, firms are divided into quartiles by value of total assets and the average debt ratios of the smallest 25 %, the largest 25 %, and those in between are reported. This panel shows that larger firms have higher long-term leverage and lower short-term leverage than smaller firms. By contrast, there do not appear to be differences in total debt ratios across firmsize quartiles. Finally, Panel C shows that, overall, public firms are less levered. The figure also shows that public firms have higher long-term and lower short-term debt ratios than private firms, but that the difference is larger in the former. This appears to be due mainly to improved profitability (as we will see momentarily) and better access to equity markets, which allows public firms to substitute the debt in their capital structure with internal and external equity, respectively. However, improved access to debt markets allows public firms to also improve the maturity of debt in their capital structure.

Examining panels A, B, and C of Fig. 1 simultaneously, we also determine a number of general themes. Specifically, we observe that regardless of firm size, industry, or stock market listing, firms' short-term leverage is much higher than their longterm leverage, implying that the majority of firms' debt is short-term. This seems to be particularly true for small private manufacturing firms. By contrast, large public manufacturing firms have the highest levels of long-term debt ratios, indicating that the usage of long-term debt is most pronounced in such firms.

¹⁸ Dividing the sample period in this way is reasonable given the fact that the Turkish economy went through a dramatic transformation following the 2001 crisis. See, for example, Turhan (2008).

¹⁹ This is, to some extent, expected as each of the industry- and macro-level factors (unlike the firm-level factors) has only two and one observations per year, respectively, and splitting the sample into two further reduces the sample size, making precise estimation of their coefficients difficult.



Fig. 1 Patterns of leverage for different types of firms. Panels $\mathbf{a}-\mathbf{c}$ of this figure display the ratios of short-term, long-term, and total debt to total assets, respectively, by industry, firm size, and stock market listing averaged over the entire sample period. In

6.1 Manufacturing firms versus non-manufacturing firms

As discussed previously in Sect. 3, industry membership may be an important determinant of firms' capital structures. In order to investigate in detail the capital structure implications of industry membership, we rerun regressions separately for manufacturing and nonmanufacturing firms. The results displayed in Table 6 suggest that manufacturing and non-manufacturing firms are in general quite similar in terms of their

"large" if it is above the third quartile capital structures. Importantly, there are no material differences in the relations between debt ratios and the determinants that we identified in Sect. 5.3 as the most

economically important such as size, profitability,

firm is classified as "small" if it is below the first quartile,

"medium" if it is between the first and third quartiles, and

tangibility, tax-related determinants, and inflation. The few small differences relate to the remaining less economically important determinants. For instance, while firm growth is positively associated with long-term leverage for manufacturing firms (consistent with the pecking order theory), it is negatively correlated with short-term leverage for
 Table 6
 Sample split between manufacturing and non-manufacturing firms

	Manufacturing firms			Non-manufacturing firms		
	Short-term leverage	Long-term leverage	Total leverage	Short-term leverage	Long-term leverage	Total leverage
Firm-specific determin	ants					
Size	1.432***	1.207***	2.615***	0.787***	0.632***	1.395***
	(0.176)	(0.134)	(0.202)	(0.123)	(0.086)	(0.138)
Profitability	-20.000***	-6.553***	-26.510***	-10.420***	-4.267***	-14.770***
	(0.900)	(0.604)	(0.995)	(1.024)	(0.698)	(1.153)
Tangibility	-0.120^{***}	0.104***	-0.017 **	-0.010	0.090***	0.080***
	(0.008)	(0.006)	(0.009)	(0.009)	(0.00709)	(0.0104)
Growth	0.0001	0.0008*	0.0010	-0.0002*	-0.0000	-0.0002
	(0.0008)	(0.0005)	(0.0007)	(0.0001)	(0.0001)	(0.0002)
Business risk	-2.646	-2.883***	-5.317***	-1.340	-5.406***	-6.559***
	(1.664)	(1.005)	(1.785)	(1.702)	(1.121)	(1.911)
Tax-related determina	nts					
Potential debt tax	43.550***	10.730***	54.160***	34.560***	10.170***	44.640***
shields	(1.772)	(1.143)	(1.910)	(2.486)	(1.614)	(2.636)
Tax exhaust	4.733***	2.366***	7.107***	3.405***	2.153***	5.583***
	(0.211)	(0.151)	(0.234)	(0.285)	(0.206)	(0.314)
Industry-specific deter	minants					
Industry median	0.421***	0.489***	0.334***	0.264***	0.426***	0.204***
leverage	(0.060)	(0.077)	(0.038)	(0.070)	(0.092)	(0.029)
Macroeconomic deteri	ninants					
Inflation	0.036***	0.019***	0.057***	0.037***	0.010**	0.056***
	(0.005)	(0.003)	(0.005)	(0.006)	(0.004)	(0.007)
GDP growth	-0.005	-0.098***	-0.081^{***}	0.013	-0.109 ***	-0.035
	(0.020)	(0.018)	(0.024)	(0.026)	(0.021)	(0.032)
Capital flows	0.046	0.075**	0.115**	0.151***	0.169***	0.210***
	(0.041)	(0.030)	(0.046)	(0.054)	(0.036)	(0.060)
Number of observations	42,998	42,998	42,998	31,134	31,134	31,134
Number of firms	6,262	6,262	6,262	5,566	5,566	5,566
Adjusted R-squared	0.646	0.577	0.658	0.657	0.654	0.692

This table presents the findings considering separately manufacturing and non-manufacturing firms. The variables are the same as those defined in Table 1 and all (except growth) are reported in three decimal places. Heteroskedasticity and serial correlation robust standard errors are reported in parentheses

***, **, and * Significance levels at the 1, 5, and 10 % levels, respectively

non-manufacturing firms (consistent with the trade-off theory). Also, comparison of the magnitudes of coefficients suggests that the debt ratios of manufacturing firms respond more to changes in size and profitability than non-manufacturing firms. The most striking difference relates to the impact of capital flows, however. In particular, the coefficient on capital flows is both much more significant and considerably larger in the non-manufacturing equations. This finding might suggest that international capital has tended to flow remarkably more to the sectors outside manufacturing, thereby influencing the financing patterns of the firms in such sectors to a greater extent during our sample period.

Overall, our findings on the relationships between leverage and various determinants appear to be more in line with the predictions of trade-off theory than with the pecking order theory, particularly in the case of non-manufacturing firms.

6.2 Large firms versus small firms

Previous research (e.g. Maksimovic and Zechner 1991, and Demirgüç-Kunt and Maksimovic 1999) provides evidence that small firms may follow financial policies that differ from those of large firms due to differences in, among other things, technology and access to financial markets and institutions. To see if this is the case, we rerun our regressions separately for the smallest and largest firms. Results displayed in Table 7 suggest that there are marked differences across firm sizes in how various determinants are related with debt ratios.

At the firm level, the most striking differences are in the effects of firm size and business risk variables. Specifically, for small firms, the association between leverage and size is relatively weak. In fact, firm size appears to become relevant for leverage decisions only when a firm is sufficiently large. This may be indicative of a "threshold effect" in size in the sense that marginal changes in the size of firms smaller than a certain threshold have little effect on such firms' ability to raise external debt. The business risk variable, on the other hand, is highly significant in all leverage regressions for small firms but always insignificant for large firms. Thus, while small firms' ability to raise debt seems very sensitive to their riskiness, large firms' access to debt is not hindered by that at all. Finally, the debt ratios of large and small firms appear to be affected differently by firm growth rates. Specifically, while firm growth is not related with leverage for small firms, there is some evidence that short-term and total leverage go down with firm growth for large firms.

Our results in Table 7 also suggest that there are interesting differences between large and small firms in how industry-related and various macroeconomic determinants are related with debt ratios. For example, the coefficient on industry median leverage is much more significant and remarkably larger for large firms, suggesting that target adjustment is a much more pertinent phenomenon among large firms. One explanation of this finding could be that there are fundamental technological differences between small and large firms and hence naturally follow different financial policies a la Maksimovic and Zechner (1991). Or, as shown by Demirgüç-Kunt and Maksimovic (1999) and World Bank Group (2010), it could reflect differences between small and large firms in access to external finance, which prevent small firms from following an optimal financial policy.

With respect to macro-level determinants, the most striking difference between large and small firm capital structures is in inflation. Specifically, while inflation does not seem to be related with the leverage decisions of small firms, large firms' leverage increases with inflation. This might suggest that, once the direct leverage impact of taxes is taken into account, inflation does not induce a further tax benefit for debt financing for small firms. Capital flows also appear to influence large and small firms differently, with capital flows having a remarkably greater importance for the leverage decisions of large firms. This might be explained by the fact that large firms are more diverse and informationally more transparent than small firms, which increases their ability to secure funds from domestic sources as well as directly from foreign sources. Finally, the association of leverage and GDP growth seems to be significantly stronger in the case of large firms, likely reflecting the fact that large firms are usually better equipped to capture the benefits of economic growth.

Overall, our findings on the relationships between leverage and various determinants appear to be more in line with the predictions of trade-off theory than with the pecking order theory, particularly in the case of large firms.

6.3 Publicly-traded firms versus private firms

Stock markets play a key role in managing conflicts of interest between various stakeholders in a firm, provide entrepreneurs with opportunities to diversify their portfolios, and transmit information about firms' prospects to potential investors and creditors (see, for example, Allen, 1993, and Demirgüç-Kunt and Table 7 Sample split between small and large firms

	Small firms			Large firms		
	Short-term leverage	Long-term leverage	Total leverage	Short-term leverage	Long-term leverage	Total leverage
Firm-specific determin	nants					
Size	0.191	0.357**	0.524*	0.774***	0.979***	1.757***
	(0.264)	(0.146)	(0.281)	(0.173)	(0.144)	(0.205)
Profitability	-12.980***	-3.992***	-17.060***	-20.560 ***	-8.258***	-28.850***
	(1.476)	(0.869)	(1.599)	(1.327)	(1.103)	(1.585)
Tangibility	-0.010	0.044***	0.033**	-0.118^{***}	0.131***	0.0127
	(0.014)	(0.009)	(0.015)	(0.0104)	(0.00983)	(0.0127)
Growth	0.0004	-0.0001	0.0002	-0.0004*	-0.0000	-0.0004**
	(0.0003)	(0.0002)	(0.0004)	(0.0002)	(0.0002)	(0.0002)
Business risk	-9.677***	-4.847***	-14.580***	0.379	-1.431	-1.033
	(2.573)	(1.400)	(2.781)	(2.384)	(1.927)	(2.773)
Tax-related determina	ints					
Potential debt tax	40.170***	1.427	41.760***	33.850***	14.800***	48.160***
shields	(4.289)	(2.204)	(4.617)	(2.395)	(1.802)	(2.652)
Tax exhaust	3.486***	0.677**	4.216***	4.447***	3.102***	7.521***
	(0.479)	(0.295)	(0.503)	(0.282)	(0.234)	(0.330)
Industry-specific deter	minants					
Industry median	0.117	0.105	0.111**	0.233***	0.574***	0.189***
leverage	(0.122)	(0.120)	(0.053)	(0.0674)	(0.117)	(0.0375)
Macroeconomic deter	minants					
Inflation	0.007	-0.004	0.007	0.067***	0.028***	0.094***
	(0.011)	(0.006)	(0.012)	(0.005)	(0.005)	(0.006)
GDP growth	0.088**	-0.063**	0.062	-0.154***	-0.118***	-0.279***
	(0.044)	(0.031)	(0.051)	(0.026)	(0.027)	(0.033)
Capital flows	0.0795	0.118**	0.151	0.131**	0.138***	0.263***
	(0.091)	(0.053)	(0.094)	(0.055)	(0.046)	(0.063)
Number of observations	14,843	14,843	14,843	20,883	20,883	20,883
Number of firms	3,456	3,456	3,456	3,245	3,245	3,245
Adjusted R squared	0.686	0.679	0.719	0.703	0.639	0.725

This table presents the findings considering separately small and large firms. Firms are divided into quartiles by value of total assets. A firm is classified as "small" if it is below the first quartile and "large" if it is above the third quartile. The variables are the same as those defined in Table 1 and all (except growth) are reported in three decimal places. Heteroskedasticity and serial correlation robust standard errors are reported in parentheses

***, **, and * Significance levels at the 1, 5, and 10 % levels, respectively

Maksimovic 1996). As a result, compared to their private counterparts, public firms enjoy improved access to financial markets in terms of both debt and

equity. This implies that whether a firm is privatelyheld or publicly-traded can have an important influence on the firm's capital structure decisions.

	Private firms			Publicly-traded firms		
	Short-term leverage	Long-term leverage	Total leverage	Short-term leverage	Long-term leverage	Total leverage
Firm-specific determin	ants					
Size	1.038***	0.843***	1.879***	2.285***	0.625	2.884***
	(0.101)	(0.074)	(0.838)	(0.844)	(0.472)	(0.838)
Profitability	-14.89***	-5.148***	-20.11***	-24.890***	-13.710***	-38.620***
	(0.687)	(0.461)	(0.766)	(3.685)	(2.539)	(4.212)
Tangibility	-0.071***	0.098***	0.027***	-0.098***	0.072***	-0.028
	(0.006)	(0.005)	(0.007)	(0.027)	(0.024)	(0.033)
Growth	-0.0002	0.0000	-0.0002	0.0039*	0.0003	0.0041
	(0.0005)	(0.0001)	(0.0002)	(0.0022)	(0.0015)	(0.0028)
Business risk	-1.798	-4.392***	-6.137***	12.26	3.336	16.23*
	(1.203)	(0.759)	(1.319)	(7.582)	(5.342)	(8.787)
Tax-related determina	nts					
Potential debt tax	39.480***	9.977***	49.340***	48.060***	19.320***	66.850***
shields	(1.498)	(0.963)	(1.606)	(6.327)	(4.012)	(6.593)
Tax exhaust	4.104***	2.240***	6.362***	7.159***	3.160***	10.280***
	(0.173)	(0.124)	(0.192)	(0.804)	(0.625)	(0.915)
Industry-specific deter	minants					
Industry median	0.350***	0.411***	0.235***	-0.466**	0.943***	0.128
leverage	(0.042)	(0.058)	(0.021)	(0.205)	(0.279)	(0.116)
Macroeconomic deterr	minants					
Inflation	0.036***	0.013***	0.055***	0.091***	0.051***	0.110***
	(0.003)	(0.002)	(0.004)	(0.015)	(0.012)	(0.015)
GDP growth	0.010	-0.111^{***}	-0.064***	-0.231***	0.100	-0.238***
	(0.058)	(0.039)	(0.060)	(0.068)	(0.070)	(0.081)
Capital flows	0.108***	0.122***	0.196***	0.128	-0.015	0.020
	(0.034)	(0.023)	(0.037)	(0.142)	(0.121)	(0.164)
Number of observations	71,885	71,885	71,885	2,270	2,270	2,270
Number of firms	11,520	11,520	11,520	206	206	206
Adjusted R squared	0.651	0.615	0.673	0.602	0.477	0.673

Table 8 Sample split between publicly-traded and private firms

This table presents the findings considering separately publicly-traded and private firms. The variables are the same as those defined in Table 1 and all (except growth) are reported in three decimal places. Heteroskedasticity and serial correlation robust standard errors are reported in parentheses

***, **, and * Significance levels at the 1, 5, and 10 % levels, respectively

In this subsection, we explore the capital structure differences between public and private firms. Toward this end, we re-run our capital structure regressions separately for public and private firms. The results are displayed in Table 8. The first thing that strikes the eye in Table 8 is that the estimated coefficients in the

private firm regressions are qualitatively and quantitatively almost the same as those in the full-sample results reported in Sect. 5. This is probably not surprising in light of the fact that nearly 98 percent of the firms in our sample are private firms.

Table 8 also shows that there are as many differences as there are similarities between public and private firms in how various determinants are related with debt ratios. At the firm level, the most important differences are in the leverage effects of firm growth and business risk. In particular, while firm growth is not related with leverage for private firms, it is positively correlated with short-term leverage for public firms, a finding consistent with the pecking order theory. Also, in stark contrast with the case for private firms, business risk appears to be positively correlated with all measures of leverage for private firms, although the correlation is significant only in the total leverage equation and only at the 10 % level. Thus, unlike the private firms, business risk does not appear to impair the ability of public firms to secure debt, most likely reflecting reputational effects and/or the presence of alternative funding sources available to public firms. Also worth noting is the fact that the coefficients on profitability are quantitatively much larger in public firm equations, suggesting that public firms can (due to higher profitability) increase the use of internal equity to a greater extent in their capital structures than private firms.

There are also interesting differences between public and private firms in how debt ratios are related with industry median leverage. In particular, while industry median leverage is positively correlated with long-term leverage for both types of firms, it is negatively correlated with short-term leverage for public firms. Therefore, it can very well be the case that improved access to financial markets in general, and debt markets in particular, may be allowing public firms to reduce short-term debt and increase long-term debt in their capital structure. This improvement in the public firms' maturity structure of debt manifests itself as a strong adjustment towards the industry median in long-term leverage coupled with a "negative" adjustment in short-term leverage.

Last but certainly not least, there is a sharp contrast between the impact of capital flows on the debt ratios of public firms and that of private firms. Specifically, while capital flows are positively correlated with leverage in all three equations for private firms, they appear to have no relation with the leverage choices of public firms. There might be a number of different explanations for this rather interesting finding. First, it might be that public firms already have nearly unlimited access to financial markets, domestic and foreign alike. If this is the case, then changes in the size and direction of capital flows would leave the capital structures of public firms unaffected. A second explanation could be that public firms are much more profitable than private firms. The availability of large internal resources, in turn, makes them less dependent on external finance and limits the impact of fluctuations in the availability of external finance on their capital structure choices.²⁰

At this point, it may be worthwhile to digress a little and summarize our findings on capital structure effects of international capital flows in light of our previous findings. We have the following facts. First, the impact of capital flows, if any, is to increase the proportion of debt (generally both short-term and long-term) in firms' capital structure. Second, the importance of capital flows for firms' capital structure decisions are considerably more apparent after the early 2000s, a period during which capital inflows to Turkey soared. Third, capital flows have a greater influence on the capital structures of (1) non-manufacturing firms compared to manufacturing firms, (2) large firms compared to small firms, (3) private firms compared to public firms, and in unreported results which are available upon request (4) mature firms than young firms. Put together, these findings suggest that the leverage-increasing impact of capital flows is most significant for large, mature, private, non-manufacturing firms after the early 2000s in Turkey. This result is in line with the findings of Falkenstein (1996) and Gompers and Metrick (2001) who find that international investors invest more in large and mature firms, and UNCTAD (2004) and Doytch and Uctum (2011) who show that in developing economies, there has been a large increase in the amount of foreign investment in services relative to that of manufacturing from the 1990s to the 2000s

Overall, our analyses of the public firm—private firm distinction suggest that the relationships between

 $^{^{20}}$ A third possibility is linked directly to our measure of capital flows, which is an aggregation of equity and debt flows. If public firms take advantage of both types of flows, this might leave their leverage unchanged.

leverage and various determinants are more in line with the predictions of trade-off theory than with the pecking order theory. However, the support for the trade-off theory is weaker and that for the pecking order theory is somewhat stronger than before in the case of public firms. To see this, note that the coefficients on firm size in the long-term leverage equation as well as on tangibility in the total leverage equation are no longer significant, which are inconsistent with the trade-off theory. Furthermore, the coefficient on firm growth in the short-term leverage equation is now positive and borderline significant, which is consistent with the pecking order theory. Finally, although the coefficient of industry median is still positive in the long-term leverage equation, it is significantly negative in the short-term leverage equation, and insignificant in the total leverage equation, findings which are not entirely consistent with the trade-off theory.

7 Pecking order or static trade-off?

Which of the two major theories, pecking order and trade-off, is a better description of our findings? Our findings suggest that, on the whole, the trade-off theory provides a better account of the capital structure decisions of Turkish firms than the pecking order theory. Table 9 presents the predictions of the two theories alongside with our empirical findings. The table includes only those determinants for which at least one of the two theories has a prediction.

A simple comparison of the signs in Table 9 reveals that the trade-off theory has more correct predictions than the pecking order theory. The main advantage of pecking order theory over the trade-off theory is that it predicts the sign of profitability correctly. This is important because profitability is probably the only leverage determinant that has a robust (negative) association with leverage in the data in (nearly) all countries. Note, however, that a negative sign can also be rationalized in dynamic versions of the trade-off theory such as Fischer et al. (1989) and Hennessy and Whited (2005). On the other hand, the pecking order theory has difficulty accounting for the positive signs on size and tangibility as well as explaining the negative association between leverage and GDP growth. Perhaps as importantly, the theory does not generate predictions about some very key leverage

 Table 9 Comparison of theoretical predictions with data facts

	Pecking order theory	Trade-off theory	Data
Firm-specific determ	ninants		
Size	_	+	+
Profitability	_	+	_
Tangibility	_	+	+
Growth	+	_	?
Business risk	-	-	-
Tax-related determi	nants		
Potential debt tax shields	?	+	+
Tax exhaust	?	+	+
Industry-specific de	terminants		
Industry median leverage	?	+	+
Macroeconomic det	erminants		
Inflation	?	+	+
GDP growth	+	_	_

This table presents the predictions of the pecking order and static trade-off theories alongside with our empirical findings for the relations between the dependent and independent variables

determinants such as corporate debt tax shields, nondebt tax shields, and inflation for which the trade-off theory correctly predicts the nature of the association with leverage.

It is important to emphasize that our conclusion is in sharp contrast with the conclusions of previous capital structure studies on the Turkish economy in which the pecking order theory is often viewed as a better framework than the trade-off theory for understanding the capital structures of Turkish firms (see, for example, Acaravcı and Doğukanlı 2004; Korkmaz et al. 2007; and Yıldız et al. 2009). We have already alluded in the preceding paragraph to some of the reasons that may at least partly explain this divergence of conclusions. Some of the other main reasons can be uncovered by studying the capital structure differences between various types of firms. This becomes all the more important since almost all of the previous studies focus on public manufacturing firms whereas we consider a comprehensive and representative sample which includes manufacturing, non-manufacturing, small, large, public, and private firms. In our analyses, we have found that the trade-off theory is particularly successful (and pecking order theory particularly unsuccessful) in accounting for the capital structures of firms that are private, non-manufacturing, large, and/or mature. Pecking order theory, on the other hand, although in our evaluation still not as powerful as the trade-off theory, is at its best when it comes to public, manufacturing, small, and/or young firms.²¹ Therefore, the reason some of the previous researchers have concluded the superiority of the pecking order theory over the trade-off theory is that they have considered only public manufacturing firms and included a very incomplete set of leverage determinants suggested by the two theories.²²

How do our findings compare with the findings on other economies? Examining the literature (some of which we discussed in Sect. 2), we see that there aren't many empirical studies that claim the superiority of the pecking order theory for developed economies. For developing economies, "votes" seem to be split roughly in half, perhaps slightly in favor of the pecking order theory. Therefore, if we had to take a very rough rule of thumb sort of stance on the issue, we could say that the trade-off theory is probably more suited to explaining the situation in developed economies, while the pecking order theory may be slightly more suited for developing economies. Several empirical studies including Deesomsak et al. (2004) and de Jong et al. (2008) show that the health and stability of the economic environment exert considerable effect on firms' capital structures. In the Turkish context, we found that in the 2003-2009 subsample where the economic environment was considerably more favorable, the predictive power of the trade-off theory was higher and that of the pecking order theory was lower compared with the case in the 1996–2002 subsample. Therefore, it might be that as the economy becomes more "healthy" and "stable", the relevance of the trade-off theory increases and that of the pecking order theory decreases. In our opinion, this is an issue that deserves further and more thorough investigation.

8 Concluding remarks

In this paper, we examine the determinants of capital structure for non-financial firms in Turkey. The novelty of our paper comes from the fact that we use a new dataset that is substantially larger and more comprehensive in terms of both time and variable coverage than those used in previous studies on individual developing economies. Our dataset includes manufacturing, non-manufacturing, small, large, listed and unlisted firms, which enables us to take a more accurate picture of the capital structure choices of the average non-financial firm as well as to analyze capital structure differences between different types of firms. Building on these comprehensive analyses, we perform hitherto the most comprehensive comparative test of the trade-off and pecking order theories in a developing economy context.

Our results provide evidence that the trade-off theory is a better framework than the pecking order theory to understand the capital structures of Turkish non-financial firms. This seems to be true regardless of firm size, industry affiliation, and stock market listing. In other words, Turkish nonfinancial firms appear to trade-off the tax benefits of debt against deadweight costs of possible bankruptcy in order to attain an optimal capital structure. The trade-off theory seems to be particularly successful in explaining the financing choices of large private non-manufacturing firms, especially when the economic environment is more stable. The pecking order theory, on the other hand, although in our assessment still not as powerful as the trade-off theory, appears to perform best for small public firms in the manufacturing sector and when the economic environment is relatively unstable.

Our conclusion that the trade-off theory is a better framework than the pecking order theory to understand the capital structures of Turkish non-financial

²¹ Our finding that pecking order theory is more powerful in explaining the financing behavior of young and small firms is consistent with notion that the pecking order theory may be more suitable for explaining the financing behavior of smaller firms since information costs are more important for SMEs than for large companies. See, for example, Zoppa and McMahon (2002) and the references therein.

²² Recall that our analysis of the economic significance of leverage determinants also lends more support to the trade-off theory than to the pecking order theory. There, we found that determinants that are more closely associated with the trade-off theory such as potential debt tax shields and firm size are economically more significant than those determinants that are more closely associated with the pecking order theory such as profitability.

firms contrasts sharply with the previous studies on the Turkish economy, who generally take the opposite view. The presence of this divergence in judgments highlights the importance of conducting sufficiently comprehensive analyses in terms of both firm types and leverage determinants suggested by various capital structure theories.

Our results also indicate that neither the trade-off theory nor the pecking order theory can match all of the observed relationships in the data. For example, a main weakness of the more successful trade-off theory is that it cannot capture the inverse relationship between leverage and profitability. As shown by Fischer et al. (1989) and Hennessy and Whited (2005), however, an inverse relationship between leverage and profitability can be rationalized in dynamic versions of the trade-off theory. This implies that dynamic versions of the theory may provide an even better framework than the static versions to think about the capital structures of non-financial firms in Turkey. A dynamic framework can also enable the analyses of the costs and speeds of adjustment in debt ratios, thereby leading to a more complete understanding of firms' target adjustment behavior than that provided in this paper. Therefore, testing the predictive ability of dynamic versions of the trade-off theory would be of great interest.

Our paper can be extended in a few other directions. First, a more in depth analysis of the leverage implications of industry affiliation would be quite valuable. In this paper, we consider two broad industry categories (manufacturing and non-manufacturing) and study leverage differences between the two. A more disaggregated industry analysis would also shed light on within-industry leverage variations as well as demonstrate whether within- or between-industry variations are the most significant. Second, our analyses in this paper provide some preliminary insights into firms' choice between short-term and long-term debt. For example, we show that large public manufacturing firms have the most favorable debt maturity structure whereas small private manufacturing firms have the least favorable. We also provide evidence of maturity matching. Our analysis, however, falls short of providing a complete analysis of the determinants of debt maturity structure, which the future research can undertake. Last but not least, given the unprecedented rise in the size and volatility of international capital flows in the past decade, it would be worthwhile to investigate in greater detail how they influence firms' capital structures.

Acknowledgments We thank Erdem Başçı for stimulating our initial interest in this topic. We also thank Cem Çebi, Mustafa Kılınç, Arif Oduncu, Baki Cem Şahin, Cihan Yalçın, Erdal Yılmaz, the editor Gerald McDermott, two anonymous referees, and seminar participants at the Central Bank of the Republic of Turkey, Turkish Economic Association's Seventeenth National Symposium on Economics in Erzurum, Borsa Istanbul Finance and Economics Conference in Istanbul, and 10th EBES Conference in Istanbul for helpful comments. We would like to also thank Aslı Kaya, Nevin Memiş, and Gülin Tuğral for kindly answering our technical questions about the dataset. The views expressed herein are solely of the authors and do not represent those of the Central Bank of the Republic of Turkey or its staff.

References

- Acaravcı, S., & Doğukanlı, H. (2004). Türkiye'de sermaye yapısını etkileyen faktörlerin imalat sanayinde sınanması. *İktisat İşletme ve Finans*, 19(225), 43–57. doi:10.3848/iif. 2004.225.3692.
- Antoniou, A., Guney, Y., & Paudyal, K. (2008). The determinants of capital structure: Capital market-oriented versus bank-oriented institutions. *Journal of Financial and Quantitative Analysis*, 43(1), 59–92. doi:10.1017/ S0022109000002751.
- Aydın, H., Kaplan, C., Kesriyeli, M., Özmen, E., Yalçın, C., & Yigit, S. (2006). Corporate sector financial structure in Turkey: A descriptive analysis. In *Central Bank of the Republic of Turkey working paper 06/07*. http://www. tcmb.gov.tr/research/discus/WP0607ENG.pdf.
- Barclay, M. J., & Smith, C. W. (1995). The maturity structure of corporate debt. *Journal of Finance*, 50(2), 609–632. doi:10.1111/j.1540-6261.1995.tb04797.x.
- Basel Committee on Banking Supervision. (2011). The transmission channels between financial and real sector: A critical survey of the literature. In *BIS working paper 18/ 11*. http://www.bis.org/publ/bcbs_wp18.pdf.
- Booth, L., Aivazian, V., Demirgüç-Kunt, A., & Maksimovic, V. (2001). Capital structures in developing countries. *Journal* of Finance, 56(1), 87–130. doi:10.1111/0022-1082.00320.
- Bradley, M., Jarrell, G. A., & Kim, E. H. (1984). On the existence of an optimal capital structure: Theory and evidence. *Journal of Finance*, 39(3), 857–878. doi:10.1111/j.1540-6261.1984.tb03680.x.
- Cebenoyan, A. S., Fischer, K. P., & Papaioannou, G. J. (1995). Corporate financial structure under inflation and financial repression: A comparative study of North American and emerging markets firms. *Global Finance Journal*, 6(1), 25–45. doi:10.1016/1044-0283(95)90010-1.
- Correa, C. A., Basso, L. F. C., & Nakamura, W. T. (2007). What determines the capital structure of the largest Brazilian firms? An empirical analysis using panel data. *Journal of International Finance & Economics*, 5(1), 27. http://connection.ebscohost.com/c/articles/31627445/

what-determines-capital-structure-largest-brazilian-firms-empirical-analysis-using-panel-data.

- De Jong, A., Kabir, R., & Nguyen, T. T. (2008). Capital structure around the world: The roles of firm and country specific determinants. *Journal of Banking & Finance*, 32(9), 1954–1969. doi:10.1016/j.jbankfin.2007.12.034.
- DeAngelo, H., & Masulis, R. W. (1980). Optimal capital structure under corporate and personal taxation. *Journal of Financial Economics*, 8(1), 3–29. doi:10.1016/0304-405X(80)90019-7.
- Deesomsak, R., Paudyal, K., & Pescetto, G. (2004). The determinants of capital structure: Evidence from the Asiapacific region. *Journal of Multinational Financial Man*agement, 14, 387–405. doi:10.1016/j.mulfin.2004.03.001.
- Dell'Ariccia, G., Di Giovanni, J., Faria, A., Kose, M. A., Mauro, P., Schindler, M., et al. (2008). Reaping the benefits of financial globalization. In *IMF occasional paper no.* 264. http://www.imf.org/external/np/res/docs/2007/0607.pdf.
- Demirgüç-Kunt, A., & Maksimovic, V. (1996). Stock market development and financing choices of firms. World Bank Economic Review, 10(2), 341–369. doi:10.1093/wber/10. 2.341.
- Demirgüç-Kunt, A., & Maksimovic, V. (1999). Institutions, financial markets, and firm debt maturity. *Journal of Financial Economics*, 54(3), 295–336. doi:10.1016/S0304-405X(99)00039-2.
- Demirhan, D. (2009). Sermaye Yapısını Etkileyen Firmaya Özgü Faktörlerin Analizi: İMKB Hizmet Firmalari Üzerine Bir Uygulama. *Ege Academic Review*, 9(2), 677–697. http://www.onlinedergi.com/MakaleDosyalari/51/PDF2009_ 2_15.pdf.
- Diamond, D. W. (1991). Debt maturity structure and liquidity risk. *Quarterly Journal of Economics*, 106(3), 709–737. doi:10.2307/2937924.
- Dinçergök, B., & Yalçıner, K. (2011). Capital structure decisions of manufacturing firms in developing countries. *Middle Eastern Finance and Economics*, 12, 86–100.
- Doytch, N., & Uctum, M. (2011). Does the worldwide shift of FDI from manufacturing to services accelerate economic growth? A GMM estimation study. *Journal of International Money and Finance*, 30(3), 410–427. doi:10.1016/j. jimonfin.2011.01.001.
- Durukan, M. B. (1997). Hisse senetleri iMKB'de işlem gören firmaların sermaye yapısı üzerine bir araştırma: 1990–1995. İMKB Dergisi, 1(3), 75–91.
- Durukan, M. B. (1998). Makroekonomik değişkenlerin türk firmalarının finansal yapısına etkisi: zaman serisi analizinden kanıtlar: 1975–1997. İktisat İşletme ve Finans, 13(147), 43–54. doi:10.3848/iif.1998.147.1276.
- Espinosa, C. M., Maquieira, C. V., Vieito, J. P., & Gonzalez, M. A. (2012). Capital structures in developing economies: The latin american case. *Investigacion Economica*, 71(282). http://www.redalyc.org/pdf/601/60125371002.pdf.
- Falkenstein, E. G. (1996). Preferences for stock characteristics as revealed by mutual fund portfolio holdings. *Journal of Finance*, 51(1), 111–135. doi:10.1111/j.1540-6261.1996. tb05204.x.
- Fischer, E. O., Heinkel, R., & Zechner, J. (1989). Dynamic recapitalization policies and the role of call premia and issue discounts. *Journal of Financial and Quantitative Analysis*, 24(4), 427–446. doi:10.2307/2330977.

- Frank, M. Z., & Goyal, V. K. (2008). Trade-off and pecking order theories of debt. In *Handbook of Empirical Corporate Finance* (pp. 135–202). Amsterdam: Elsevier Science. doi:10.1016/B978-0-444-53265-7.50004-4.
- Frank, M. Z., & Goyal, V. K. (2009). Capital structure decisions: Which factors are reliably important? *Financial Management*, 38(1), 1–37. doi:10.1111/j.1755-053X.2009.01026.x.
- Gompers, P. A., & Metrick, A. (2001). Institutional investors and equity prices. *Quarterly Journal of Economics*, 116(1), 229–259. doi:10.1162/003355301556392.
- Gönenç, H. (2003). Capital structure decisions under micro institutional settings: The case of Turkey. *Journal of Emerging Market Finance*, 2(1), 57–82. doi:10.1177/ 097265270300200103.
- Harris, M., & Raviv, A. (1991). The theory of capital structure. *Journal of Finance*, 46(1), 297–355. doi:10.1111/j.1540-6261.1991.tb03753.x.
- Hennessy, C. A., & Whited, T. M. (2005). Debt dynamics. Journal of Finance, 60(3), 1129–1165. doi:10.1111/j. 1540-6261.2005.00758.x
- Hovakimian, A., Opler, T., & Titman, S. (2001). The debtequity choice. *Journal of Financial and Quantitative Analysis*, 36, 1–24. doi:10.2307/2676195.
- Huang, G., & Song, F. M. (2006). The determinants of capital structure: Evidence from China. *China Economic Review*, 17(1), 14–36. doi:10.1016/j.chieco.2005.02.007.
- Korkmaz, T., Albayrak, A. S., & Karataş, A. (2007). Hisse senetleri İMKB'de işlem gören kOBİ'lerin sermaye yapısının incelenmesi: 1997–2004 Dönemi. İktisat İşletme ve Finans, 22(253), 79–96. doi:10.3848/iif.2007.253. 1548.
- Kose, M. Ayhan, Prasad, E., Rogoff, K., & Wei, S. J. (2009). Financial globalization: A reappraisal. *IMF Staff Papers*, 56(1), 8–62. doi:10.1057/imfsp.2008.36.
- Kraus, A., & Litzenberger, R. H. (1973). A state-preference model of optimal financial leverage. *Journal of Finance*, 33, 911–922. doi:10.1111/j.1540-6261.1973.tb01415.x.
- Mackay, P., & Phillips, G. M. (2005). How does industry affect firm financial structure? *Review of Financial Studies*, 18, 1433–1466. doi:10.1093/rfs/hhi032.
- Maksimovic, V., & Zechner, J. (1991). Debt, agency costs, and industry equilibrium. *Journal of Finance*, 46, 1619–1643. doi:10.1111/j.1540-6261.1991.tb04637.x.
- Miller, M. (1977). Debt and taxes. *Journal of Finance*, *32*, 261–275. doi:10.1111/j.1540-6261.1977.tb03267.x.
- Mishkin, F. S. (1997). The causes and propagation of financial instability: Lessons for policymakers. In: Hakkio, C. (Ed.), *Maintaining Financial stability in a global economy* (pp. 55–96). Kansas City: Federal Reserve Bank of Kansas City. https://www.kansascityfed.org/publicat/sympos/ 1997/pdf/s97mishk.pdf.
- Modigliani, F., & Miller, M. H. (1958). The cost of capital, corporation finance and the theory of investment. *American Economic Review*, 48(3), 261–297. http://www.jstor.org/ stable/1809766.
- Myers, S. C. (1984). The capital structure puzzle. *Journal of Finance*, *39*(3), 75–92. doi:10.1111/j.1540-6261.1984. tb03646.x.
- Myers, S. C. (2003). Financing of corporations. *Handbook of the Economics of Finance* (pp. 215–253). Amsterdam: North Holland/Elsevier.

- Myers, S. C., & Majluf, N. S. (1984). Corporate financing and investment decisions when firms have information that investors do not have. *Journal of Financial Economics*, *13*(2), 187–221. doi:10.1016/0304-405X(84)90023-0.
- Newey, W. K., & West, K. D. (1987). A simple, positive semidefinite, heteroskedasticity and autocorrelation consistent variance matrix. *Econometrica*, 55, 703–708. www.jstor. org/stable/1913610.
- Okuyan, H. A., & Taşçı, H. M. (2010). Sermaye yapısının belirleyicileri: türkiye'deki en büyük 1000 sanayi işletmesinde bir uygulama. *Journal of BRSA Banking and Financial Markets*, 4(1), 105–120. http://www.bddk.org.tr/ WebSitesi/english/Reports/BRSA_Journal/81575.makale %20ing.pdf.
- Ostry, J. D., Ghosh, A. R., Chamon, M., & Qureshi, M. S. (2011). Capital controls: When and why? *IMF Economic Review*, 59(3), 562–580. doi:10.1057/imfer.2011.15.
- Pandey, I. M. (2004). Capital structure, profitability and market structure: Evidence from Malaysia. Asia Pacific Journal of Economics & Business, 8(2), 78–91.
- Pozdena, R. J. (1987). Tax policy and corporate capital structure. *Economic Review* (Fall), 37–51. http://www.frbsf.org/ economic-research/files/87-4_37-51.pdf.
- Qureshi, M. A. (2009). Does pecking order theory explain leverage behaviour in Pakistan? *Applied Financial Economics*, 19(17), 1365–1370. doi:10.1080/0960310090 2817592.
- Rajan, R. G., & Zingales, L. (1995). What do we know about capital structure? Some evidence from international data. *Journal of Finance*, 50(5), 1421–1460. doi:10.1111/j. 1540-6261.1995.tb05184.x.
- Reinhart, C. M., & Rogoff, K. S. (2011). From financial crash to debt crisis. *American Economic Review*, 101(5), 1676–1706. doi:10.1257/aer.101.5.1676.
- Sayılgan, G., Karabacak, H., and Küçükkocaoğlu, G. (2006). The firm-specific determinants of corporate capital structure: Evidence from Turkish panel data. *Investment Man*agement and Financial Innovations, 3(3), 125–139. http:// www.baskent.edu.tr/~gurayk/kisiselcapstrpaper.pdf.
- Schmukler, S. L., & Vesperoni, E. (2006). Financial globalization and debt maturity in emerging economies. *Journal* of Development Economics, 79(1), 183–207. doi:10.1016/j. jdeveco.2004.12.006.
- Schoubben, F., & Van Hulle, C. (2004). The determinants of leverage: Differences between quoted and non-quoted firms. *Tijdschrift voor Economie en Management*, 49(4).

https://lirias.kuleuven.be/bitstream/123456789/274828/2/ tem_4-04_04_schoubben.pdf.

- Shuetrim, G., Lowe, P., and Morling, S. (1993). The determinants of corporate leverage: A panel data analysis. *Reserve Bank of Australia working paper no: 9313*. http://www.rba. gov.au/publications/rdp/1993/pdf/rdp9313.pdf.
- Stohs, M. H., & Mauer, D. C. (1996). The determinants of corporate debt maturity structure. *Journal of Business*, 69(3), 279–312. http://www.jstor.org/stable/2353370.
- Taggart, Jr. R. A. (1985). Secular patterns in the financing of US corporations. In *Corporate capital structures in the United States* (pp. 13–80). Chicago: University of Chicago Press. http://www.nber.org/chapters/c11417.pdf.
- Titman, S., & Wessels, R. (1988). The determinants of capital structure choice. *Journal of Finance*, 43(1), 1–19. doi:10. 1111/j.1540-6261.1988.tb02585.x.
- Turhan, I. M. (2008). Why did it work this time: A comparative analysis of transformation of Turkish economy after 2002. *Asian-African Journal of Economics and Econometrics*, 8, 255–280.
- UNCTAD. (2004). *The shift toward services*. New York and Geneva: United Nations. http://unctad.org/en/Docs/ wir2004_en.pdf.
- Wald, J. K. (1999). How firm characteristics affect capital structure: An international comparison. *Journal of Financial Research*, 22(2), 161–187. http://papers.ssrn.com/ sol3/papers.cfm?abstract_id=142576.
- Wiwattanakantang, Y. (1999). An empirical study on the determinants of the capital structure of Thai firms. *Pacific-Basin Finance Journal*, 7(3), 371–403. doi:10.1016/ S0927-538X(99)00007-4.
- World Bank Group (2010). From crisis to private sector led growth. In *Investment climate assessment (Turkey)*. Washington, DC. http://documents.worldbank.org/ curated/en/2010/05/12786352/turkey-investment-climateassessment-crisis-private-sector-led-growth.
- Yıldız, M. E., Yalama, A., & Sevil, G. (2009). Sermaye Yapısı Teorilerinin Geçerliliğinin Test Edilmesi: Panel Veri Analizi Kullanılarak İMKB-İmalat Sektörü Üzerinde Ampirik Bir Uygulama. İktisat İşletme ve Finans, 24(278), 25–45. doi:10.3848/iif.2009.278.4175.
- Zoppa, A., & McMahon, R. G. (2002). Pecking order theory and the financial structure of manufacturing SMEs from Australia's business longitudinal survey. *Small Enterprise Research*, 10(2), 23–42. doi:10.5172/ser.10.2.23.